Author's response to reviews

Title: Rationale and design of the randomised clinical trial comparing early medication change (EMC) strategy with treatment as usual (TAU) in patients with Major Depressive Disorder - the EMC trial

Authors:

André Tadic (tadic@uni-mainz.de)
Stanislav Gorbulev (gorbulev@izks-mainz.de)
Norbert Dahmen (ndahmen@uni-mainz.de)
Christoph Hiemke (hiemke@uni-mainz.de)
Dieter F. Braus (dbraus@uke.de)
Joachim Röschke (roeschke@scivias-caritas.de)
Dietrich van Calker (dietrich.calker@uniklinik-freiburg.de)
Daniel Wachtlin (wachtlin@izks-mainz.de)
Kai Kronfeld (kronfeld@izks-mainz.de)
Thorsten Gorbauch (gorbauch@izks-mainz.de)
Monika Seibert-Grafe (seibert-grafe@izks-mainz.de)
Klaus Lieb (klaus.lieb@uniklinik-freiburg.de)

Version: 3 Date: 25 February 2010

Author's response to reviews:

25 February 2010

Dear Madams, dear Sirs,

please find attached our revised Ms. 1845477624325058 by Tadi# et al. As requested, we deleted reference n° 31. As a consequence, references n° 32-43 of the MS version R1 are now named n° 31-42, resp. Additionally, we changed the format of those text passages, which indicated changes between the original MS and version R1 (from "italics and underlined" into normal text). During the process of resubmission, we deleted the additional material files ("ethics approval" and "funding"), because these files were confidentially provided to the editors and are not for publication.

We hope that you will find our revised manuscript now fully suitable for publication.

With best regards,

André Tadic, MD
(Corresponding author)